Author's response to reviews

Title: Deficiency of G1 regulators P53, P21Cip1 and/or pRB decreases hepatocyte sensitivity to TGFbeta cell cycle arrest

Authors:

Sharon Sheahan (sheahan@hotmail.com)
Christopher O Bellamy (c.bellamy@ed.ac.uk)
Donald R Dunbar (donald.dunbar@ed.ac.uk)
David J Harrison (david.Harrison@ed.ac.uk)
Sandrine Prost (s.prost@ed.ac.uk)

Version: 5 Date: 19 November 2007

Author's response to reviews:

Reference to figure 5 has now been added - Reference to Figure 4 has also been changed as it was to figure 4A rather than figure 4.

I hope this is all OK now

cheers
S Prost